**Author’s response to reviews**

**Title:** LncRNA PLAC 2 downregulates miR-21 in non-small cell lung cancer and predicted survival

**Authors:**

Huan Xia (himysc567632@126.com)
Ming Xiu (elprdlnn7376611@126.com)
Jinying Gao (qfbdoh2761137@126.com)
Hongyu Jing (sehqcbvh221756471@126.com)

**Version:** 2  **Date:** 14 Jul 2019

**Author’s response to reviews:**

Dear Editor,

Thanks for your suggestions! We have revised the manuscript according to your requirements.

Editor Comments:

1. In your “Ethics approval and consent to participate” section of your Declarations please clearly state whether the consent obtained from the participants was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

   RE: It is Written informed consent, Revised.

2. Please represent authors' names using their initials, not their full name, in the Authors’ Contributions section of your Declarations. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

   RE: Revised.

3. Thank you for uploading your editing certificate. As this file is no longer required at this stage of the editorial process, please remove it from your submission.
RE: Revised.

4. Please upload your revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. Please also ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

RE: Revised.

Best wishes!

Hongyu Jing